Author's response to reviews

Title: Understanding healthcare practices in superdiverse neighbourhoods and developing the concept of welfare bricolage. Protocol of a cross-national mixed-methods study

Authors:

Jenny Phillimore (j.a.phillimore@bham.ac.uk)
Hannah Bradby (hannah.bradby@soc.uu.se)
Michi Knecht (knecht@uni-bremen.de)
Beatriz Padilla (beatrice.padilla@iscte.pt)
Tilman Brand (brand@bips.uni-bremen.de)
Sin Yi Cheung (CheungSY@cardiff.ac.uk)
Simon Pemberton (s.pemberton@keele.ac.uk)
Hajo Zeeb (zeeb@bips.uni-bremen.de)

Version: 3 Date: 5 June 2015

Author's response to reviews:

Dear Diorelle, dear Editor,

we have revised the manuscript as suggested, giving the full e-mail information on the title page and moving the table as indicated. Let me once again confirm that this study proposal was peer-reviewed in a competitive two-stage procedure, in the framework of the NORFACE call under which it is now being funded.

Greetings

Hajo Zeeb